Author's response to reviews

Title: Analysis of surgical and oncological Outcome in internal and external Hemipelvectomy in 34 Patients above the Age of 65 Years at a mean Follow-up of 56 Months

Authors:

   Wiebke K Guder (wiebke.guder@ukmuenster.de)
   Jendrik Hardes (jendrik.hardes@ukmuenster.de)
   Georg Gosheger (georg.gosheger@ukmuenster.de)
   Marcel-Philipp Henrichs (marcel-philipp.henrichs@ukmuenster.de)
   Markus Nottrott (markus.nottrott@ukmuenster.de)
   Arne Streitbürger (streitb@ukmuenster.de)

Version: 2 Date: 30 July 2014

Author's response to reviews:

Dear Ladies and Gentlemen,

according to your request, we have revised our manuscript as follows:

1. Email addresses of all authors were included at Title Page.
2. We improved the format of our abstract and main text document according to the guidelines for authors.
3. Acknowledgements: Due to retrospective study design, no other contributors - but the authors - were involved in the process of preparing the uploaded manuscript. Therefore, we didn't include an Acknowledgements section in the uploaded document. All authors meet the criteria for authorship.

   If you would like us to proceed differently regarding the Acknowledgments section (i.e. state the no acknowledgements need to be made), please let us know and we'll meet your requirements.

Requesting Ethics Statement: Due to the retrospective study design, the approval of an Ethics committee wasn't necessary for carrying out our analysis. We included a sub-section Ethics Approval within the Methods section stating that no approval needed to be obtained.

We hope that you'll find that our document now meets your requirements to be considered for peer review.

We're looking forward to your reply.

Sincerely,
Wiebke Guder